Author's response to reviews

Title: A case of squamous cell carcinoma arising from a suprapubic cystostomy tract

Authors:

Hiroki Ito (pug_daikichi@yahoo.co.jp)
Masayuki Arao (tanntann.com@gmail.com)
Hanako Ishigaki (hanako.9.11@hotmail.co.jp)
Noboru Ohshima (noboruswk@gmail.com)
Ayako Horita (mainichi-daikichi@ezweb.ne.jp)
Ikuo Saito (sleepynaoki@yahoo.co.jp)
Kotaro Hirai (k-hirai@sagamihara-hosp.gr.jp)

Version: 5 Date: 29 September 2011

Author's response to reviews: see over
“A case of squamous cell carcinoma arising from the suprapubic cystostomy tract”

Ito H*1, Arao M*1, Ishigaki H*1, Ohshima N*2, Horita A*3, Saito I*3, Hirai K*1.

National Hospital Organization Sagamihara National Hospital
*1 Department of Urology   *2 Department of Dermatology    *3 Department of Pathology

Dear Editors

Thank you for acceptance for your journal.

We had made any correction to our article based on opinion pointed out in major revision section.

Sincerely yours,
Hiroki Ito